

Received 7 August 2023, accepted 25 August 2023, date of publication 30 August 2023, date of current version 19 September 2023. Digital Object Identifier 10.1109/ACCESS.2023.3310238

RESEARCH ARTICLE

Massive MIMO Detectors Based on Deep Learning, Stair Matrix, and Approximate Matrix Inversion Methods

MAHMOUD A. ALBREEM[®], (Senior Member, IEEE), KHAWLA A. ALNAJJAR[®], (Member, IEEE), AND ALI J. ALMASADEH[®]

Department of Electrical Engineering, University of Sharjah, Sharjah, United Arab Emirates

Corresponding author: Khawla A. Alnajjar (kalnajjar@sharjah.ac.ae)

This work was supported in part by the Seed Research Project under Grant 22020403209, and in part by the Competitive Research Project of the University of Sharjah under Grant 23020403252.

ABSTRACT Massive multiple-input multiple-output (MIMO) is an essential technology in fifth-generation (5G) and beyond 5G (B5G) communication systems. Massive MIMO is employed to meet the increasing request for high capacity in next-generation wireless communication networks. However, signal processing in massive MIMO incurs a high complexity due to a large number of transmitting and receiving antenna elements. In this paper, we propose low complexity massive MIMO data detection techniques based on zero-forcing (ZF) and vertical bell laboratories layered space-time (V-BLAST) method in combination with approximate matrix inversion techniques; Neumann series (NS) and Newton iteration (NI). The proposed techniques reduce the complexity of the ZF V-BLAST method since they avoid the exact matrix inverse computation. Initialization based on a stair matrix is also exploited to balance the performance and the complexity. In addition, we propose a massive MIMO detector based on approximate matrix inversion with a stair matrix initialization and deep learning (DL) based detector; MM Network (MMNet) algorithm. MMNet contains a linear transformation followed by a non-linear denoising stage. As signals propagate through the MMNet, the noise distribution at the input of the denoiser stages approaches a Gaussian distribution, form precisely the conditions in which the denoisers can attenuate noise maximally. We validated the performance of the proposed massive MIMO detection schemes in Gaussian and realistic channel models, i.e., Quadriga channels models. Simulations demonstrate that the proposed detectors achieve a remarkable improvement in the performance with a notable computational complexity reduction when compared to conventional ZF V-BLAST and the MMNET in both simple and real channel scenarios.

INDEX TERMS 5G, massive MIMO, Gauss-Seidel, successive overrelaxation, Neumann series, Newton iteration, deep learning, data detection, stair matrix, diagonal matrix, V-BLAST.

I. INTRODUCTION

Massive multiple-input multiple-output (MIMO) is a crucial technology in developing next-generation wireless communication networks, as it offers numerous advantages, such as increased network capacity and improved reliability. Over the past decade, a large number of transmitting and receiving antennas have been employed in massive MIMO networks to support the rapid increase in demand for bandwidth. Moreover, a large number of antennas in massive MIMO

The associate editor coordinating the review of this manuscript and approving it for publication was Chengpeng Hao^(D).

systems improve spectral efficiency and reliability, which are critical requirements for beyond fifth-generation (B5G) wireless networks. However, the computational complexity of massive MIMO detectors increases when a large number of antennas are utilized, making the conventional data detection techniques impractical for large-scale MIMO configurations. A survey of massive MIMO detectors is shown in [1] and [2], where a wide range of detectors are illustrated and criticized. The maximum likelihood (ML) based detector achieves the best performance [3]. Unfortunately, ML based detector uses an exhaustive search which causes the complexity to grow exponentially as the number of antennas increases in the

This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License. For more information, see https://creativecommons.org/licenses/by-nc-nd/4.0/

massive MIMO systems [3]. Linear and non-linear massive MIMO detection techniques are studied to find near-optimal performance while reducing the computational complexity. Linear detectors, including the matched filter (MF) [4], the zero-forcing (ZF) [5], and the minimum mean-square error (MMSE) [6] are dependent on the matrix inversion, which is a computationally expensive mathematical operation in massive MIMO systems. Various iterative methods that use matrix inverse approximation are proposed to achieve good performance with low complexity when compared with MMSE based detector [7]. Using approximate matrix inversion in these detectors reduces the complexity required by avoiding the computation of the exact matrix inverse of the Gramian matrix [8]. These methods work well under specific configurations for massive MIMO detection, i.e., the number of antenna elements at the base station (BS) is larger than the number of transmitting users [1]. Neumann series (NS) is a popular matrix inverse approximation method implemented to reduce the complexity compared to the MMSE based detector [9]. The performance of the NS techniques suffers when the ratio of the receiving antennas and users is close to 1. Newton iteration (NI) is another iterative method that can obtain an approximate matrix inverse with good precision using a few iterations [10].

It is well known that the equalization matrix is diagonally dominant. Therefore, most existing detectors are utilizing the diagonal matrix in their detection. However, detection methods based on a diagonal matrix may not converge under certain circumstances, especially when the number of users approaches the number of BS antennas. In [11], stair matrices are discussed with a focus on their applications to iterative methods. Moreover, initialization is developed based on a stair matrix and is exploited for massive MIMO detectors through approximate matrix inversion methods. Different linear massive MIMO detectors, such as the Gauss-Seidel (GS) and the successive over-relaxation (SOR), are proposed to achieve a performance gain compared to other approximate matrix inversion methods [12].

However, linear detectors may suffer from a performance loss when the number of users approaches the number of BS antennas; hence, non-linear detectors are proposed. The vertical bell laboratories layered space-time (V-BLAST) is a non-linear detection algorithm based on linear detectors. The detection algorithm detects the transmitted symbols iteratively and utilizes linear detection techniques at each iteration [13].

A. RELATED WORK

Deep learning (DL) is one of the most auspicious technologies for artificial intelligence (AI) and achieved enormous success in signal and image processing, speech recognition, and biomedical sciences. The DL approach uses artificial neural networks to solve complex problems by modeling these problems into layers of neurons that are inspired by the human brain [14]. In wireless communication systems, it can be trained effectively to predict/approximate the transmitted signal vectors [15]. DL based detectors can be classified as: the single channel realization architectures and varying channel realization architectures [2]. The DL based detectors were used to attain an optimal performance while minimizing the complexity. However, the results of the detection techniques degrade when dealing with real channel models and high modulation schemes. Additionally, some of the DL techniques suffer from high computational complexity in specific scenarios preventing their implementation in real applications. Significant research in recent years used the DL approach such as [3], [16], [17], [18], [19], [20], and [21].

Since 2017, there has been a prominent trend in the research and industry communities to employ DL in massive MIMO detectors. DL is recruited to shift the computational complexity of the massive MIMO detectors to the offline phase, enabling faster run-time in the online data detection phase. The deep network in a massive MIMO detector is employed in DetNet [22] where a projected gradient descent method is used. DetNet performs well in simple channel scenarios and with low-order modulation schemes. A modified DetNet requires a small number of parameters to be optimized; unfortunately, DetNet training is unstable in the case of realistic and correlated channels. In addition, the scalability of the DetNet algorithm is miserable because of a huge number of training parameters. In recent years (2018-2023), a substantial stream in the research sector to avail DL to create a robust massive MIMO detector is observed.

A model-driven DL network is demonstrated based on the orthogonal approximate message passing network (OAMP-Net) [23]. The detector based on OAMP-Net adds some adjustable parameters to the existing OAMP method, but a strict assumption has to exist. The OAMP-Net is dominated by the matrix inverse in each layer. Hence, it is not feasible for real-life applications/implementation due to high computational complexity. The OAMP-Net performance is improved, and OAMP-Net2 is proposed in [24], where new training parameters are demonstrated. Unlike the OAMP-Net, imperfect channels are taken into consideration. However, similar to the original OAMP-Net, it is dominated by the matrix inverse. The MMNet algorithm is proposed to conquer challenges in both the DetNet and the OAMPNet [25]. MMNet is designed to be trained online for each channel matrix and an iterative soft thresholding algorithm is used. Although it obtains a satisfactory performance when implemented in a realistic channel simulator, MMNet incurs high latency due to sequential online training. In addition, the performance significantly deteriorates when high modulation schemes are employed.

A HyperMIMO based detector displaces the MMNet training process by a single inference over a trained hyper-network where the number of MMNet parameters are reduced [26]. In comparison with the MMNet, HyperMIMO performs slightly worse. It also has to be re-trained when serious changes in the channel statistics are observed. In [27], a massive MIMO detector based on an efficient data-driven DL

TABLE 1. Advantages and disadvantages of existing DL based detectors.

DL Technique	Advantages	Disadvantages		
DetNet [22]	 It has adequate performance when tested on constant and Rayleigh fading channels. It performs well in low-order modulation such as BPSK and 4-QAM. 	 It contains a large number of trainable parameters, which makes the technique un-scalable for high-order modulation and large MIMO systems. It uses a single detection model for different channel models. The performance is not stable for realistic and correlated channels. 		
OAMP-Net [23]	 It achieves higher SER performance than DetNet on i.i.d. Gaussian channels. OAMP-Net solves the issue of the high number of trainable parameters that exist in other DL algorithms. OAMP-Net uses only two trainable parameters in each layer and performs well while using low- order modulation schemes. 	 It experiences high complexity because of the matrix inverse operation required in each layer. The performance is degraded in realistic and correlated channels. It uses a single detection model for different channel models and does not perform well in massive MIMO systems. 		
MMNet [25]	 It addresses the bad performance of OAMPNet and DetNet in realistic channels. It offers online training for each realization of <i>H</i> and does not use matrix inversion in each layer. It has lower complexity than OAMPNet and DetNet architectures. It has better performance than the MMSE, DetNet, and OAMPNet in realistic channels. It provides, in the linear and denoising stages of each layer, a balance between flexibility and complexity. 	 The performance is reduced when using high-order modulation schemes. It requires the neural network to be retained for each realization of the channel matrix (<i>H</i>), which causes the real practical implementation to be complicated. It experiences latency because of the sequential online training. 		
HyperMIMO [32]	 It has lower complexity than the MMNet. This is realized by minimizing the number of trainable parameters in each layer. It has superior performance compared to the OAMP-Net DL algorithm. It was practically implemented and has good robustness against user mobility. 	 It has a lower performance than the MMNet algorithm. It requires to be retained when the channel matrix is changed considerably. 		
DLNet [16], [17]	 A new loss function is proposed based on the sum of mean squared error between the transmitted and estimated signals at each layer. It avoids matrix inversion. It can be used in a variety of channel distributions and statistics. 	• It is not tested in real channel scenarios.		

network is proposed. An extrapolation factor is exploited as a learnable parameter for the iterative sequential detector. Moreover, additional learnable parameters and multiple softsign activation functions are demonstrated to host different high modulation orders.

A model-driven DL structure for data detection and channel estimation is proposed in [28]. The proposed special DL structure takes advantage of domain knowledge in the few-bit quantization process and relies on the original quantized system model. Unfortunately, the performance of this detector significantly deteriorates when spatially correlated channels are used. In [16], a massive MIMO decoder based on DL with 40 layers has been proposed. At each layer, a loss function is demonstrated based on the sum of mean-squared errors between transmitted and estimated signals. This decoder achieves a significant computational complexity reduction due to the non-existence of matrix inversion. However, this method is not tested in the case of realistic channel scenarios. In [29], a ZF-ML framework with DL approach is utilized to design an efficient massive MIMO detector. The DL is employed to suppress the interfering signals. The work in [30] considers the DL approach for a small-scale MIMO

detectors and shows the benefits of DL employment in the performance-complexity profile of MIMO detector.

The work in [15] provides insight into how to leverage the DL scheme for massive MIMO detector design. In [31], approximate matrix inversion methods are exploited to enhance the performance of the MMNet while attaining low computational complexity. The utilization of approximate matrix inversion methods such as the GS and the SOR with MMNet method provided remarkable performance enhancement for the MMNet detection architecture [31].

Table 1 presents a brief overview of the DL-based detectors such as the DetNet, the OAMP-Net, the MMNet, and the HyperMIMO. It is clear that most existing DL based detectors are unstable for realistic channel scenarios or suffer a significant performance deterioration compared to the ML based detectors. However, they perform well in Gaussian channels and low modulation schemes. Motivated by this challenge, we revisit large-scale MIMO detection to strike a balance between performance and complexity in realistic channel scenarios.

B. CONTRIBUTION AND ORGANIZATION

Massive MIMO technology is employed in 5G communication systems to achieve a high quality-of-service (QoS) and increase capacity. However, massive MIMO receiver design is not a trivial task. One of the major problems in massive MIMO is achieving a satisfactory trade-off in the detector's performance-complexity profile. This challenge is becoming imperious when the number of transmitting antennas is approaching the number of receiving antennas. In addition, this problem is more substantial in the existence of a practical channel scenario. The motivation for this paper originates from obtaining an attractive trade-off between the computational complexity and the performance of massive MIMO detectors based on DL and approximate matrix inversion methods in realistic channel scenarios.

This paper also introduces a low-complexity massive MIMO detection method based on a modified version of the V-BLAST algorithm and the approximate matrix inversion methods such as the NS and NI. It should be noted that the initialization stage has a great impact on the convergence rate, and hence the detectors performance and computational complexity. Therefore, the stair matrix initialization is employed for both the NS and NI methods to improve the performance of the proposed V-BLAST massive MIMO detector. Design of the proposed algorithm ensures that good performance is maintained and a considerable reduction in complexity is achieved since matrix inversion is avoided in the ZF V-BLAST algorithm by using proposed iterative methods. Moreover, a stair matrix initialization with approximate matrix inversion is leveraged to improve the performance of a DL based detector.

The main contributions of the paper are summarized as follows:

 We propose massive MIMO detection techniques that have good performance and low complexity based on ZF V-BLAST and the approximate matrix inversion techniques, such as the NS and NI methods. The proposed techniques avoid exact matrix inversion in the ZF V-BLAST algorithm by using iterative methods. Moreover, the stair matrix initialization is utilized for both NS and NI methods to reduce the computational complexity of the proposed detectors with a high-performance gain.

- We also propose a hybrid structure between the DL and approximate matrix inversion methods. The initialization stage is also developed based on the stair matrix structure while the detection process exploits the MMNet structure.
- We use extensive simulations to investigate the performance and the computational complexity of the proposed massive MIMO detectors in different configurations. In order to avoid any misleading conclusion, perfect channel state information (CSI) and real channel scenarios are considered.

The rest of the paper is organized as follows. Section II introduces the massive MIMO system model. In Section III, we discuss different relevant linear and non-linear massive MIMO detection methods. Section IV describes the proposed massive MIMO detection technique. Section V provides the complexity analysis. Section VI presents the implementation details and performance results of the proposed techniques. Finally, summary and future research directions are provided in Section VII.

Notation: This list demonstrates the notations which are used in the paper:

a, μ, A	Scalar. Italics characters of standard weight.
v	Vector. Boldface, lower case.
М	Matrix. Boldface, upper case alphabetical.
\boldsymbol{v}_i	Element i of vector v .
$\boldsymbol{M}_{i,j}$	Element at row <i>i</i> and column <i>j</i> of matrix,
	or matrix expression, M .
[M] _{mm}	The $(m, m)^{\text{th}}$ element of a matrix M .
$\boldsymbol{M}^T, \boldsymbol{v}^T$	Matrix and vector transpose.
M^{-1}	Matrix inverse.
H^+	Moore-Penrose generalized inverse of <i>H</i> .
$\ v\ $	Euclidean norm of vector \boldsymbol{v} .
$\{a_i\}$	The set containing elements a_i .
$max\{\cdot\}$	The maximum value taken by the argument.
$\min\{\cdot\}$	The minimum value taken by the argument.
E[x]	Expectation of random variable x.
I _K	$K \times K$ identity matrix.

II. SYSTEM MODEL

The system model for the detection of uplink massive MIMO system is provided in (1) [32]. We assume a massive MIMO system with N_r total number of receiving antennas at the BS and N_t users with a single antenna. The transmitted symbols by the transmitters are considered as a $N_t \times 1$ vector denoted by \mathbf{x} while the received signal is a $N_r \times 1$ vector represented by \mathbf{y} . The relationship between the received signal

y and the transmitted symbols x is described as

$$\boldsymbol{y} = \boldsymbol{H}\boldsymbol{x} + \boldsymbol{n},\tag{1}$$

where **n** is $N_r \times 1$ noise vector that is assumed as additive white Gaussian noise (AWGN) $n_i \sim C\mathcal{N}(0, \sigma^2)$ with independent and identically distributed (i.i.d) entries.

The channel matrix (H) represents the channel coefficients between the N_t single antenna users and the N_r receiving antennas at the (BS). The channel matrix (H) for an $N_r \times N_t$ massive MIMO system is represented as

$$\boldsymbol{H} = \begin{bmatrix} h_{11} & h_{12} & \cdots & h_{1j} \\ h_{21} & h_{22} & \cdots & h_{2j} \\ \vdots & \vdots & \vdots & \vdots \\ h_{i1} & h_{i2} & \cdots & h_{ij} \end{bmatrix}.$$
 (2)

The power for the channel coefficient h_{ij} between the *j*th user and *i*th receiving antenna at the BS is provided by $P_{ij} = E\left[|h_{ij}|^2\right]$. The noise power is given by $E\left[|n_i|^2\right] = \sigma^2$ and transmitted signals are normalized where $E\left[|x_j|^2\right] = 1$. The transmitted symbols **x** are selected randomly from the predefined constellation set by the transmitting antennas. The signal-to-noise ratio (SNR) is represented by SNR = $E\left[|x_j|^2\right]/\sigma^2$.

Gaussian and realistic channel models, in addition to channels with imperfect channel estimation, are considered to evaluate the proposed detection techniques. The Gaussian channels assume variables that have zero mean and unity variance. Moreover, the channel coefficients are considered to be i.i.d complex Gaussian random variables in addition to having a power spectral density (PSD) of $N_0/2$ [33]. In the case of realistic channels, we consider the quasi-deterministic radio channel generator (Quadriga) model provided in [34].

The Quadriga simulator is utilized for the simulation of realistic channels in massive urban MIMO system configurations. The Quadriga channel model based on the 3rd generation partnership project (3GPP) can generate realistic channel impulse responses that consider various factors such as the users' variable speeds, arbitrary length of the channel traces, and multi-dimensional propagation [34]. We also consider the model presented in [35] for the imperfect channel estimation. In practical systems, the receiver has an estimate \hat{H} of the actual channel H. The channel state information (CSI) for the imperfect channel is given by $\hat{H} = r_0 H + \sqrt{1 - r_0^2 E}$, [35] where r_0 represents the correlation coefficient between the actual channel H and the estimated channel \hat{H} . The error matrix E is statistically identical to H and is independent of H.

III. MASSIVE MIMO DETECTION

A. CLASSICAL LINEAR MASSIVE MIMO DETECTORS

The purpose of massive MIMO detection is to determine the transmitted symbols vector sent by the users using the received signal and the channel matrix [36]. The maximum likelihood (ML) achieves the best performance. However, it performs an exhaustive search to find all the possible solutions for the received signal. The ML algorithm is represented as [3]

$$\hat{\boldsymbol{x}}_{ML} = \underset{\boldsymbol{x} \in O^{K}}{\arg\min} \|\boldsymbol{y} - \boldsymbol{H}\|_{2}^{2}, \qquad (3)$$

where \hat{x}_{ML} is received signal estimated by the ML detector. However, the computational complexity of the ML based detector increases exponentially as the number of antennas increases O^{N_t} [3]. Therefore, the ML algorithm is unsuitable for practical implementation due to a high computational complexity.

1) LINEAR ZF DETECTOR

The ZF detector has better performance than the MF detector [5]. It tries to maximize the received signal-to-interference-plus-noise ratio (SINR) by inverting the channel matrix. The equalization matrix for linear ZF detector is given as [5]

$$\boldsymbol{A}_{ZF}^{H} = \left(\boldsymbol{H}^{H}\boldsymbol{H}\right)^{-1}\boldsymbol{H}^{H} = \boldsymbol{H}^{+}, \qquad (4)$$

where H^+ represent the Moore-Penrose pseudo-inverse of the matrix H. It should be noted that the ZF detector ignores the noise and works well when the interferences are limited.

2) LINEAR MMSE DETECTOR

Minimum mean-square estimation (MMSE) is a linear detection technique that takes the effects of the noise into consideration. Hence, it outperforms the MF and ZF linear detectors at lower SNR [37]. It works by minimizing the mean-square error between the transmitted signal (x) and $H^H y$ which is presented as [37]

$$\boldsymbol{A}_{\text{MMSE}}^{H} = \arg\min_{\boldsymbol{H}\in^{N_{r}\times N_{t}}} E \left\|\boldsymbol{x} - \boldsymbol{H}^{H}\boldsymbol{y}\right\|^{2}.$$
 (5)

The MMSE detector takes the effects of noise into consideration and is represented as [6]

$$\boldsymbol{A}_{\text{MMSE}}^{H} = \left(\boldsymbol{H}^{H}\boldsymbol{H} + \sigma^{2}\boldsymbol{I}\right)^{-1}\boldsymbol{H}^{H}.$$
 (6)

The MMSE has high computational complexity since it depends on high-order matrix inversion. Moreover, the algorithm suffers from unstable performance for higher-order matrices [6]. Additionally, the performance of the MMSE based detector is severely degraded when working under ill-conditioned channels [2].

B. LINEAR DETECTORS BASED ON APPROXIMATE MATRIX INVERSION

1) NEUMANN SERIES

The NS is an approximate matrix inversion method used to reduce the complexity of linear detection [8]. The Gramian matrix $G = H^H H$ inversion is a highly computational task. The NS simplifies the matrix inversion by decomposing the matrix G into G = D + E, where D is the diagonal matrix, and E is presenting the non-diagonal elements. The Gramian matrix inversion based on the NS method can be represented as [8]

$$\boldsymbol{G}^{-1} = \sum_{n=0}^{\infty} \left(-\boldsymbol{D}^{-1} \boldsymbol{E} \right)^n \boldsymbol{D}^{-1}, \tag{7}$$

where the condition for convergence of the NS algorithm to the matrix inverse is provided by [8]

$$\lim_{n \to \infty} \left(-\boldsymbol{D}^{-1} \boldsymbol{E} \right)^n = 0.$$
(8)

It should be noted that the NS method suffers from a considerable performance loss when the ratio of the number of receiving antennas to users is close to 1. When the number of iterations is low $(n \le 2)$, the NS method has a computational complexity of $O(N_t^2)$.

2) NEWTON ITERATION METHOD

The NI method, also called the Newton-Raphson method, is an iterative method used to find the approximate matrix inverse [38]. The estimate of the matrix inverse at a specific iteration (n) is provided as [38]

$$\boldsymbol{X}_{n}^{-1} = \boldsymbol{X}_{n-1}^{-1} \left(2\boldsymbol{I} - \boldsymbol{G} \boldsymbol{X}_{n-1}^{-1} \right).$$
(9)

The NI method was shown to converge faster when compared to the NS method, even though it requires one additional multiplication in each iteration [10]. The NI method has a computational complexity of $O(N_t^2)$ [7].

3) GAUSS-SEIDEL METHOD

The GS method is an iterative technique that is employed to estimate the matrix inversion, and it has been employed in linear detectors [39]. The Gram matrix G can be decomposed as G = D + U + L, where U is the upper triangular matrix of G, and L is the lower triangular matrix of G. The detected signal for the GS detector is estimated as [40]

$$\hat{\boldsymbol{x}}^{(n)} = (\boldsymbol{D} + \boldsymbol{L})^{-1} \left(\hat{\boldsymbol{x}}_{MF} - \boldsymbol{U} \hat{\boldsymbol{x}}^{(n-1)} \right).$$
(10)

The computational complexity for the GS based detector is $O(N_t^2)$ [40]. The GS detector's initial estimate could be considered a zero vector if it is unknown [41]. However, a low convergence rate is noticed when zero vector is used for initialization.

4) SUCCESSIVE OVER-RELAXATION METHOD

A relaxation factor ω can be used to improve the convergence and performance of the GS method. The SOR method has a $O(N_t^2)$ computational complexity that is similar to the GS method [42]. The GS is considered as a special case of the SOR method where $\omega = 1$ [43]. The SOR is reliant on having a suitable value for the relaxation parameter. Since the prediction of the optimal relaxation parameter is a challenging task, performance degradation could occur if an inappropriate value of ω is used. The estimated signal for the SOR detector is given as [42]

$$\hat{\boldsymbol{x}}^{(n)} = \left(\frac{1}{\omega}\boldsymbol{D} + \boldsymbol{L}\right)^{-1} \left(\hat{\boldsymbol{x}}_{\mathrm{MF}} + \left((\frac{1}{\omega} - 1)\boldsymbol{D} - \boldsymbol{U}\right)\hat{\boldsymbol{x}}^{(n-1)}\right).$$
(11)

It should be noted that all above-mentioned iterative methods are convergent for any initial solution $(\hat{x}^{(0)})$ if the the equalization matrix is strictly diagonally dominant (SDD). However, the convergence rate is related to the spectral radius of the iterative method. Hence, each iterative method has it is own convergence rate. Moreover, the convergence rate depends on the selection of $\hat{x}^{(0)}$ [12]. In [11] and [44], the relationship between the convergence rate, maximum eigenvalue, and minimum eigenvalue is comprehensively discussed. It is also shown that the convergence conditions can be satisfied when the number of BS antennas is sufficiently large. Table 2 summarizes the pros and cons of conventional massive MIMO detectors.

5) STAIR MATRIX INITIALIZATION FOR APPROXIMATE MATRIX INVERSION DETECTORS

The performance, complexity, and convergence rate of the approximate matrix inversion methods used in linear massive MIMO detection is highly dependent on the initial values. The equalization matrix is considered to be diagonally dominant in MMSE based detector. Hence, most of the existing approximate matrix inverse iterative detectors use the diagonal matrix. However, the solution of the linear iterative methods may not converge when the diagonal matrix is adopted [44]. The stair matrix is a special case of the tridiagonal matrix that has zero off-diagonal elements in either the odd or even rows [11]. Tridiagonal matrix can be classified as a stair matrix (S) if it satisfies one of the following conditions [12]:

$$S_{(i,i-1)} = 0,$$

$$S_{(i,i+1)} = 0, \quad i = 2, 4, \dots, 2 \left| \frac{N_t}{2} \right|.$$

$$S_{(i,i-1)} = 0,$$

(12)

$$\mathbf{S}_{(i,i+1)} = 0, \quad i = 1, 3, \dots, 2 \left| \frac{N_t - 1}{2} \right| + 1.$$
 (13)

For example, a stair matrix with a dimension of 6×6 can be represented in the following form [12]:

In [12], the stair matrix is exploited instead of the diagonal matrix to initialize several approximate matrix inversion massive MIMO detectors. Most approximate matrix inversion methods utilize $\hat{x}_{(0)} = D^{-1}y_{MF}$ for selection of the

TABLE 2. Pros and cons of conventional massive MIMO detection techniques.

Technique	Pros	Cons
MF	It works well when the users count is much lower than the receiving antennas [4]. It has low complexity [4].	The method exhibits bad performance in ill-conditioned channels [4].
ZF	It has low complexity [5]. It has good performance in environments where inter- ference is limited [5].	It has bad performance in ill-conditioned channels [2] [5].
MMSE	It has better performance than MF and ZF techniques [37]. It considers the noise effects and reduces noise enhancements.	It has bad performance in ill-conditioned channels. The MMSE has high complexity and suffers from un- stable performance for higher-order matrices [6].
GS	The technique performs well when β is close to 1.	The algorithm is difficult for parallel implementation because it is reliant on sequential iterations [45].
SOR	The technique performs well when β is close to 1. The algorithm managed to outperform NS and CG- based detectors [43].	It is dependent on the relaxation parameter ω that is difficult to predict [1].
CG	It has satisfactory performance when the ratio of receiv- ing antennas and users is high [46]. The CG can provide numerically stable performance [47].	The number of operations is high [46]. The method needs a large number of iterations [46].
NS	It has low complexity [8].	It has bad performance when β is close to 1 [8]. The algorithm is slower compared with NI method [38].
NI	It has a faster convergence rate than NS [38].	It requires a higher number of multiplications than NS [38].
Local search	It can achieve near-optimal performance by using the fixed neighborhood to minimize the ML cost [48].	It has prohibitive complexity and reduced performance when using high modulation schemes [48].
Belief propaga- tion	It can attain near-optimal results in cases where the number of antennas is high and with low correlated channels [49].	This iterative technique does not always ensure the solution will converge [49]. It is dependent on finding the optimal damping factor which is difficult to predict [49].
BOX Detection	It can achieve satisfactory performance with inexpensive hardware complexity [50].	The performance suffers when the ratio of antennas at BS and users is high [51].
V-BLAST (SIC)	It has superior performance compared to ZF and MMSE techniques [33].	The method requires more complexity compared to ZF and MMSE [33]. The results of the algorithm are influenced by the first detected signal [33].

initial solution. The detector based on approximate matrix inversion methods such as the GS and SOR with stair matrix initialization is discussed in [12]. The algorithm uses the stair matrix to obtain the initial solution that is given by $\hat{x}_{(0)} = S^{-1}y_{MF}$ for the GS and SOR linear iterative methods. The operators diag(A), triu(A), tril(A), and stair(A) is used to represent the commands used to obtain the diagonal, upper triangular, lower triangular, and stair matrix components of A, respectively. In addition, massive MIMO detectors based on NI and NS with stair matrix are proposed in [12]. The GS and SOR methods achieved the best performance with the lowest complexity compared to NS and NI based detectors [12]. Moreover, the complexity involved in computing S^{-1} and D^{-1} is the same, with both operations having a computational cost of $O(N_t)$ [12].

C. VERTICAL BELL LABORATORIES LAYERED SPACE-TIME (V-BLAST) ALGORITHM

V-BLAST is a non-linear detection technique that is based on linear detectors such as the linear ZF detector. It is also called order successive interference cancellation (OSIC). The V-BLAST detection algorithm is shown in [32]. The interference introduced by the different antennas is canceled using the V-BLAST algorithm. The algorithm performs the detection and the cancellation sequentially in a serial fashion. First, it detects, using the ZF detector, the signal with the highest SINR to ensure the best performance. After that, the algorithm detects the second strongest signal and arranges for the cancellation of the signal from the remaining signal set. The algorithm is repeated iteratively until the detection of all the signals is completed. The ZF V-BLAST algorithm has better performance than classical linear detectors such as the ZF. However, the ZF V-BLAST method has higher computational complexity [32]. The two main factors that increase the computational complexity of the ZF V-BLAST algorithms are the ordering and cancellation process in addition to the matrix inversion computation for the equalization matrix A [32].

In the following, we compare the effect of interference of linear detectors such as ZF and maximum ratio combiner

(MRC) and compare them with VBLAST. Following [52], the SINR for the m^{th} user is

$$SINR_{m} = \frac{[|\vec{w}_{m}^{H}h_{m}x_{m}|^{2}]}{[|\vec{w}_{m}^{H}(\sum_{j\neq m}\vec{h}_{j}x_{j}+\vec{n})|^{2}]}$$
(15)

where \vec{w}_m is the m^{th} column of the linear detecting matrix, W.

With perfect CSI, ZF removes the interference while interference increases for MRC. As discussed in [53], the sum of the interference is required to be small before the MRC receiver is similar to the ZF receiver because the aggregate interference corrupts MRC while ZF deletes it. In this paper, we assume self-interference is negligible because each user has a single antenna. ZF, in general, suppresses the effect of interference and suffers from noise enhancement under the assumption of the perfect channel. However, VBLAST is based on successive interference cancelation and detection, so if there are no other users, we come back to the case of the linear detector, and hence if we use ZF VBLAST, we will have a match of the ZF case [53]. Note that if we assume, each user has multiple antenna and we have multiple users, self-interference will occur in case no other users appears. This can be explored for future work.

D. MMNet DEEP LEARNING ARCHITECTURE

The MMNet is a massive MIMO DL network architecture with a varying channel realization architecture [25]. The MMNet addresses the poor performance of the DetNet and the OAMP-Net in realistic channel models. The MMNet architecture balances flexibility and complexity within each layer of the neural network. Earlier DL detection techniques such as the DetNet and OAMP-Net are trained offline and have single channel realization architecture. However, the MMNet is designed with varying channel realization and supports both online and offline training. For the MMNet, training and testing data are generated through the massive MIMO model illustrated in (1) where the signal (\mathbf{x}) , the channel matrix (H) and the noise (n) are the sources of randomness. In addition, each x is generated randomly and uniformly over the corresponding constellation set. Adam optimizer is utilized with a learning rate of 10^{-3} . In order to have a fair comparison, we used the training parameters and hyperparameters as mentioned in [25]. The MMNet architecture that is implemented for realistic channels has ten layers with $2N_t(N_r+1)$ trainable parameters per layer. For arbitrary channels, the MMNet architecture model is designed to be suitable as in [25]

where

$$\hat{\boldsymbol{x}}_{t+1} = \eta_t \left(\boldsymbol{z}_t; \sigma_t^2 \right). \tag{17}$$

(16)

The MMNet has a complex trainable matrix ($\Theta_t^{(1)}$) that enables the model to work in scenarios where several noise levels influence the transmitted signal. This enables the MMNet algorithm to perform well in realistic environments

 $\boldsymbol{z}_t = \hat{\boldsymbol{x}}_t + \boldsymbol{\Theta}_t^{(1)} \left(\boldsymbol{y} - \boldsymbol{H} \hat{\boldsymbol{x}}_t \right),$



FIGURE 1. Block diagram of the MMNet architecture [25].

with arbitrary channels. Fig. 1 shows the block diagram for the MMNet architecture. MMNet neural network contains a linear transformation followed by a non-linear denoising stage. It repeatedly refines an estimate of the signal by alternating between a linear detection stage and a non-linear denoising stage. When signals propagate through the MMNet stages, the noise distribution at the input of the non-linear denoiser stages approaches a Gaussian distribution, creating precisely the conditions in which the denoisers can attenuate noise maximally. An efficient massive MIMO detection technique has been proposed based on the initialization of the MMNet algorithm using several approximate matrix inversion methods such as GS, SOR, and conjugate gradient (CG) [31]. The massive MIMO detector based on the initialization of the MMNet with the GS method achieved the best performance, followed by SOR and CG. Moreover, all proposed detectors significantly improve performance compared to MMNet while maintaining low complexity [31]. The loss function of MMNet is calculated as Loss = $\frac{1}{L}\sum_{l=1}^{L} \|\hat{\boldsymbol{x}}_{l} - \boldsymbol{x}\|_{2}^{2}$ where L presents the number of layers.

IV. PROPOSED MASSIVE MIMO DETECTOR

The ZF V-BLAST algorithm outperforms linear detection techniques such as ZF for massive MIMO systems. However, the ZF V-BLAST algorithm is highly complex due to the matrix inversion, which is computationally demanding. We propose a low-complexity massive MIMO detection algorithm based on the linear ZF detection method and the V-BLAST algorithm. The ZF V-BLAST algorithm is enhanced by using two iterative approximate matrix inversion methods: the NS and NI.

Matrix inversion is used to sustain good performance while reducing computational complexity for massive MIMO detection. In Fig. 2, we provide the block diagram of the proposed massive MIMO detection technique based on the low complexity V-BLAST algorithm. In the proposed technique, the exact matrix inverse in the iterative process of the ZF V-BLAST algorithm is replaced with matrix inverse approximation using NS and NI methods.

One of the major bottlenecks for the ZF V-BLAST algorithm is the computational complexity required to calculate the exact matrix inverse for the Gramian matrix. We consider two approximate matrix inversion methods, the NS and NI, for estimating the matrix inverse. The equations for the NS and NI methods to avoid the calculation of matrix inverse are provided in (7) and (9), respectively. We use a small



FIGURE 2. Block diagram for the proposed low complexity V-BLAST massive MIMO detection algorithm.

number of iterations ($n \le 2$) for the NS and NI approximate matrix inversion methods in order to ensure that the computational complexity remains low for the proposed algorithms. Algorithm 4 shows the proposed massive MIMO detection technique.

In the proposed algorithm, the Gramian matrix [54] is obtained first for ZF linear detector, which is given by $B_{ZF} = H^H H$. The matrix inverse is approximated using the NS and NI methods. The number of iterations is kept equal to or less than 2 in order to ensure that the complexity of the approximate methods is $O(N_t^2)$. The approximated matrix inverse is then used in the equalization matrix for the classical linear detection techniques. The algorithm detects the signal with the highest SINR using ZF to ensure the best performance. After that, the second strongest signal is detected and arranged for the cancellation of the signal from the remaining signal set. The optimal ordering for a ZF V-BLAST algorithm is given by [32]

$$k_i^{(\text{ZF})} = \arg\min_{j \notin \{k_1, k_2, \dots, k_{i-1}\}} \|(\boldsymbol{A}_i)_j\|^2, \qquad (18)$$

where $(\mathbf{A}_i)_j$ is the j^{th} of \mathbf{A} at the i^{th} iteration. The algorithm is repeated for each iteration until detecting all the signals is completed.

V. COMPLEXITY ANALYSIS

We investigate the computational complexity of the proposed massive MIMO detection techniques in this section. The computational complexity is dominant by the number of complex multiplications. The computational complexity for different massive MIMO detection algorithms in terms of O(.) notation is presented in Table 3. We refer to the proposed detection techniques as ZF V-BLAST NS and ZF V-BLAST NI when the NS and NI are used, respectively, to approximate the matrix inverse for the ZF V-BLAST algorithm.

Using V-BLAST	Algorithm and Approximate Matrix			
Inversion Methods				
1) Initialization:				
i=1	<i>i</i> is the iteration number.			
Н	H is the channel matrix.			
у	y is the received signal.			
2) Ordering and				
successive				
cancellation:				
В	Find Gramian matrix for the ZF linear			
	detector.			
B _{ZF}	For ZF, $\boldsymbol{B}_{\boldsymbol{ZF}} = \boldsymbol{H}^H \boldsymbol{H}$.			
B^{-1}	Use NS and NI to obtain the			
	approximate matrix inverse.			
	For NS, $\mathbf{B}^{-1} = \sum_{n=0}^{\infty} (-\mathbf{D}^{-1}\mathbf{E})^n \mathbf{D}^{-1}$.			
	For NI, $\boldsymbol{X}_n^{-1} = \boldsymbol{X}_{n-1}^{-1} \left(2\boldsymbol{I} - \boldsymbol{B} \boldsymbol{X}_{n-1}^{-1} \right).$			
Α	Evaluate the linear detector.			
	For ZF, $\boldsymbol{A}_{ZF} = (\boldsymbol{B}_{ZF})^{-1} \boldsymbol{H}^{H}$.			
$\boldsymbol{G} = \boldsymbol{A}^H$	Obtain the Hermitian of the linear			
	detector, A .			
k_i	Obtain the selection index k_i .			
	For ZF, $k_i = \arg \min \ (\mathbf{A}_i)_i \ ^2$.			
	$j \notin \{k_1, k_2,, k_{i-1}\}$			
$\boldsymbol{m}_{k_i} = (\boldsymbol{G}_i)_{k_i}$	\boldsymbol{m}_{k_i} is the i^{th} row of \boldsymbol{G} .			
$\tilde{x}_i = \boldsymbol{m}_{k_i} \boldsymbol{y}$	Evaluate the estimated input signal \tilde{x}_i .			
$\hat{x}_i = \hat{Q}[\tilde{x}_i]$	$\hat{Q}(.)$ is the quantization operation			
	based on the used constellation.			
$\mathbf{y}_{i+1} = \mathbf{y}_i - \mathbf{h}_{k_i} \hat{x}_{k_i}$	The interference due to \hat{x}_{k_i} is cancelled.			
$\boldsymbol{H}_{i+1} = \boldsymbol{H}_i^{\overline{k_i}}$	Update H at iteration <i>i</i> by zeroing the			
·	k_i column.			
i = i + 1	Update <i>i</i> .			

Algorithm 1 Proposed Massive MIMO Detection Technique

The computational complexity for matrix inverse using the Gaussian elimination method is provided by $O(ab^2)$ for an $a \times b$ matrix [55]. In the ZF V-BLAST algorithm, it is required to obtain the matrix inverse of $H^H H$ for the ZF

MIMO detection technique	Complexity		
ZF [57]	$O(N_r N_t^2)$		
ZF V-BLAST [53]	$O(N_r N_t^3)$		
ZF V-BLAST NS	$O(N_r N_t^3)$		
ZF V-BLAST NI	$O(N_r N_t^3)$		
MMNet [25]	$O(bN_r^2L)$		
MMNet-GS	$O(bN_r^2L)$		
MMNet-SOR	$O(bN_r^2L)$		

 TABLE 3. Comparison of the computational complexity order for different linear and non-linear massive MIMO detection techniques.

techniques. Hence, the matrix inverse in the ZF V-BLAST algorithm has a computational complexity of $O(N_t^3)$. Moreover, the same complexity order of $O(N_t^3)$ is required if Cholesky decomposition is utilized to compute the matrix inverse [56]. The matrix inversion causes high complexity for the ZF V-BLAST algorithm. The proposed technique reduces the complexity of the matrix inverse operation in the original ZF V-BLAST algorithm by avoiding the matrix inversion computation using the NS and NI methods which have a matrix inverse complexity of $O(N_t^2)$. Hence, matrix inversion complexity is reduced from $O(N_t^3)$ to $O(N_t^2)$. ZF V-BLAST and the proposed detection techniques have a computation complexity of $O(N_r N_t^3)$. However, the proposed detection algorithms that use the NS and NI approximate matrix inversion methods can lower the computational complexity by reducing the number of multiplications required by the ZF V-BLAST algorithm. Hence, the computational complexity required for matrix inversion using NS and NI methods is $O(N_t^2)$ when the number of iterations is equal to or less than 2 $(n \leq 2).$

The number of multiplications required for each massive MIMO detection technique is summarized in Table 4. In the conventional ZF V-BLAST algorithm, $\frac{N_r N_t (N_t+1)}{2}$ multiplications are required in order to obtain $\boldsymbol{H}^{H}\boldsymbol{H}$ and $\frac{N_{t}^{2}}{2} + \frac{3}{2}N_{t}^{2}$ multiplications are required in order to find $(\mathbf{H}^H \mathbf{H})^{-1}$ [53], [56]. For example, the number of multiplications required at the first stage of the ZF V-BLAST algorithm is provided by $\frac{N_rN_t(N_t+1)}{2} + \frac{N_t^3}{2} + \frac{3}{2}N_t^2 + N_t^2N_r + N_tN_r + 2N_r$. The same can be obtained for the second and third stages of the ZF V-BLAST algorithm by substituting $(N_t - 1)$ and $(N_t - 2)$, respectively. Hence, the number of multiplications required for the ZF V-BLAST algorithm is provided by $\frac{N_r \sum_{i=0}^{N_t - 2} (N_t - i)(N_t - i + 1)}{2} + \frac{\sum_{i=0}^{N_t - 2} (N_t - i)^3}{2} + \frac{3}{2} \sum_{i=0}^{N_t - 2} (N_t - i)^2 + \sum_{i=0}^{N_t - 2} (N_t - i)^2 N_r + (N_t N_r + 2N_r (N_t - 1) + N_r)$ [32]. For the NS and NI methods, the number of multiplications required to obtain the matrix inverse is $N_t^2 + Nt$ when the number of iterations is $n \leq 2$. Hence, the term required to obtain the matrix inverse $\frac{N_t^3}{2} + \frac{3}{2}N_t^2$ is reduced $N_t^2 + Nt$. For example, the number of multiplications required to obtain the exact matrix inverse for a massive MIMO system with $N_t = 32$ is 17920 multiplications. While using the NS and NI methods would require 1056 multiplications to approximate the matrix inverse. Hence, the number of multiplication required to obtain the matrix inverse is reduced considerably. It is also worth noting that the computational complexity of the MMNet [25] can be presented as $O(bN_r^2L)$.

A. COMPLEXITY OF STAIR MATRIX IN APPROXIMATE MATRIX INVERSION METHODS

The number of the real multiplications required to obtain the inverse of the stair matrix is $3(N_t - 1)$. In comparison, the number of division operations required to obtain the inverse of the diagonal matrix is N_t . Although the initialization of the approximate matrix inversion methods with stair matrix increases the number of multiplications to $3(N_t - 1)$, this increase in complexity is considered negligible. For example, the NS detector with a stair matrix requires 17453 multiplications while the conventional NS detector with a diagonal matrix requires 17408 for a 16×64 massive MIMO system and n = 2. Consequently, the computational complexity to obtain S^{-1} is $O(N_t)$, which is equivalent to the computational complexity required to calculate D^{-1} . Table 4 shows the number of multiplications required with and without stair matrix initialization for the proposed ZF V-BLAST NS, ZF V-BLAST NI, MMNet-GS, and MMNet-SOR detection techniques. It is shown that the increase in the number of multiplications by $3(N_t - 1)$ in the proposed algorithms is insignificant. Hence, the exploitation of the stair matrix initialization does not affect the complexity order for the proposed detection techniques.

In order to have a fair comparison between the proposed detection techniques and the recent work in [16], Table 5 illustrates the required number of flops and the estimated execution time when the computer is executing 1 million operation per second. A 8×64 MIMO system is employed where each complex multiplication requires 3 flops. It is clear that the work in [16] has the lowest execution time. However, the proposed detection techniques still have a very low execution time.

VI. SIMULATION RESULTS

The performance results of the proposed massive MIMO detection techniques based on the linear ZF detector and the V-BLAST algorithm in conjunction with approximate matrix inversion methods are presented in this section. We implement the proposed massive MIMO detection algorithms using two different architectures (ZF V-BLAST NS and ZF V-BLAST NI). We first compare the performance of the proposed detection techniques using the symbol error rate (SER) versus SNR. The main performance criteria will be the SNR required to achieve a range of $10^{-2} - 10^{-3}$ SER since, typically, the error correction schemes operate in this range for MIMO detection [25].

The ZF V-BLAST algorithm is implemented using the NS and NI methods for the inverse matrix approximation.

Detection technique	Complexity
ZF [53]	$0.5N_t \left(N_t^2 + 3N_t \left(N_r + 1 \right) + 3N_r \right)$
ZF V-BLAST [33]	$\frac{\frac{N_r \sum_{i=0}^{N_t - 2} (N_t - i)(N_t - i + 1)}{2}}{2} + \frac{\frac{\sum_{i=0}^{N_t - 2} (N_t - i)^3}{2}}{2} + \frac{3}{2} \sum_{i=0}^{N_t - 2} (N_t - i)^2 + \sum_{i=0}^{N_t - 2} (N_t - i)^2 N_r + (N_t N_r + 2N_r (N_t - 1) + N_r)}$
ZF V-BLAST NS/NI	$\frac{N_r \sum_{i=0}^{N_t - 2} (N_t - i)(N_t - i + 1)}{2} + \sum_{i=0}^{N_t - 2} (N_t - i)^2 + \sum_{i=0}^{N_t - 2} (N_t - i)^2 N_r + (N_t N_r + 2N_r (N_t - 1) + N_r)$
ZF V-BLAST NS/NI with stair matrix	$\frac{\frac{N_r \sum_{i=0}^{N_t - 2} (N_t - i)(N_t - i + 1)}{2} + \sum_{i=0}^{N_t - 2} (N_t - i)^2 + \sum_{i=0}^{N_t - 2} (N_t - i)^2 N_r + (N_t N_r + 2N_r (N_t - 1) + N_r) + 3(N_t - 1)}{N_r}$
MMNet-GS	GS $(n = 1)$: $(4N_r + 2)N_t^2 + 2(N_r - 1)N_t$
MMNet-SOR	SOR $(n = 1)$: $(4N_r + 2)N_t^2 + 2(N_r)N_t$
MMNet-GS with stair matrix	GS $(n = 1)$: $(4N_r + 2)N_t^2 + 2(N_r - 1)N_t + 3(N_t - 1)$
MMNet-SOR with stair matrix	SOR $(n = 1)$: $(4N_r + 2)N_t^2 + 2(N_r)N_t + 3(N_t - 1)$
DLNet [16], [21]	$4N_t^3 + 9N_t^2 - 5N_t$

TABLE 4. Computational complexity of the proposed detection techniques and linear detection methods.

TABLE 5. Required number of flops and execution time if 8×64 MIMO system is employed.

Method	Number of flops	Execution time	
MMNet-GS	52560	0.05256	
MMNet-SOR	52608	0.052608	
MMNet-GS with stair ma- trix	52623	0.052623	
MMNet-SOR with stair matrix	52671	0.05261	
DLNet [16]	7752	7.752×10^{-3}	



FIGURE 3. Performance comparison for different ZF V-BLAST massive MIMO detection algorithms using 8 × 64 MIMO system.

We refer to the ZF V-BLAST implemented with NS and NI as ZF V-BLAST NS and ZF V-BLAST NI, respectively. We test the ZF V-BLAST NS and ZF V-BLAST NI proposed techniques on different massive MIMO system configurations. In the first scenario, a system size ratio size N_t/N_r of 0.125 and 0.25 for $N_r = 64$. In the second scenario, the same system size ratios are tested for BS with $N_r = 32$.



using i.i.d. Gaussian channels. For a system size of 8×64 , the proposed ZF V-BLAST NS and ZF V-BLAST NI detection techniques. As shown in Fig. 3 and Fig. 4, both of the proposed ZF V-BLAST NS and ZF V-BLAST NI detection techniques have attained excellent performance in an 8×64 massive MIMO system while having lower complexity than the original ZF V-BLAST algorithm. Similarly, the proposed detection techniques achieved satisfactory performance in the 16×64 massive MIMO system. It should be noted that ZF V-BLAST NI has better performance than ZF V-BLAST NS in massive MIMO systems with $N_r = 64$.



FIGURE 4. Performance comparison for different ZF V-BLAST massive MIMO detection algorithms using 16 × 64 MIMO system.

The detection techniques are tested in the third scenario for massive MIMO configurations with 0.125 and 0.25 massive MIMO system size ratios with $N_r = 128$. The modulation scheme used in the simulation is (quadrature amplitude modulation) QAM-4, with each simulation repeated for 100,000 - 1,000,000 experiments. For brevity, the results for the first scenario are only shown in the paper.

In the second scenario, the ZF V-BLAST NS, ZF V-BLAST NI, and ZF V-BLAST with exact matrix inversion are tested using i.i.d. Gaussian channels. The two massive MIMO system configurations considered for this scenario are 4×32 and 8×32 system sizes. The proposed ZF V-BLAST NS and ZF V-BLAST NI detection techniques achieve satisfactory SER performance in comparison to the exact ZF V-BLAST method while having lower complexity. It was also noticed that when the number of users increased to $N_t = 8$, the ZF V-BLAST NS and ZF V-BLAST NI still managed to achieve good performance compared with the original ZF V-BLAST algorithm with exact matrix inversion. Moreover, the ZF V-BLAST NI detector attains superior performance when compared to the ZF V-BLAST NS detector in both 4×32 and 8×32 massive MIMO system configurations. In the third scenario, the ZF V-BLAST NS, ZF V-BLAST NI, and ZF V-BLAST with exact matrix inversion are tested using i.i.d. Gaussian channels. The ZF V-BLAST NS and ZF V-BLAST NI achieved similar results to the first and second scenarios for both 16×128 and 32×128 system sizes. Both ZF V-BLAST NS and ZF V-BLAST NI algorithms had reliable SER with significantly lower complexity than the original ZF V-BLAST algorithm with exact matrix inverse for the mentioned system sizes. It should be noted that as the ratio between the transmitting users and the receiving antennas increases, the performance of the detection techniques will start to suffer. The reduction in performance is expected as the system is loaded with more users since this is the typical behavior for the ZF V-BLAST, NS, and NI methods. Accordingly, the proposed ZF V-BLAST NS and ZF V-BLAST NI massive MIMO detection techniques can achieve the best performance in massive MIMO systems where the number of receiving antennas is much larger than the number of users.

The simulation shows that the proposed detection techniques, which are ZF V-BLAST NS and ZF V-BLAST NI, have good performance in massive MIMO scenarios where the number of receiving antennas is much larger than the number of users. Moreover, the proposed detection techniques have excellent performance results that are similar to the original ZF V-BLAST algorithm when the system size ratio N_t/N_r is 0.125. In the case where the system size has a ratio of 0.25, the proposed detection techniques achieved SER of around 10^{-3} within a reasonable SNR.

A. STAIR MATRIX INITIALIZATION FOR THE PROPOSED MASSIVE MIMO DETECTION TECHNIQUES

The performance of the proposed massive MIMO detection techniques when implemented with the stair matrix initialization is presented in this section. In the first scenario, the ZF V-BLAST NS and ZF V-BLAST NI algorithms are implemented with a stair matrix and diagonal matrix for approximate matrix inversion methods. The ZF V-BLAST, ZF V-BLAST NS, and ZF V-BLAST NI algorithms are simulated using i.i.d. Gaussian channel model with massive MIMO system size of 8×32 . The number of iterations used for both the NI and NS methods is 2. As shown in



FIGURE 5. Performance comparison for different ZF V-BLAST massive MIMO detection algorithms with stair matrix initialization using 8 \times 32 MIMO system.

Fig. 5, the ZF V-BLAST NS and ZF V-BLAST NI algorithms implemented with stair matrix achieve better results than ZF V-BLAST NS and ZF V-BLAST NI algorithms implemented with the diagonal matrix, respectively.

As discussed in [53], the SER is controlled by the cases when all the users have weak link powers. For instance, the interference sum is small. All users have weak link powers at low SNR; hence the total interference is slight. At higher SNR and in the early V-BLAST stages, the interference can be quite significant since the users do not have weak link powers. The most substantial user leads and faces a high SINR. On the other hand, this low SER scenario does not significantly influence the overall SER. In the later V-BLAST stages, the users face a lower SINR, and the involvement of the total SER is greater.

In the second scenario, the GS and SOR methods are utilized with a stair matrix using 1 iteration for the initialization of MMNet in offline training mode. The training of the MMNet model was carried out using a batch size of 500 samples, with 10000 iterations, and a test batch size of 5000 samples. We evaluate the performance of the algorithms using Quadiga channels for a massive MIMO system with $N_t = 8$ users and $N_r = 64$ antennas at the BS. Moreover, we assume a modulation scheme of 4-QAM in the simulation. The results for MMNet in addition to MMNet-GS, and MMNet-SOR initialized with diagonal and stair matrices are shown in Fig. 6a. Moreover, a comparison is provided for the performance of the MMNet-GS detector with stair matrix initialization using Gaussian and Quadriga channels in Fig. 6b. The results show that the realistic channels are significantly more challenging than Gaussian channels.

Table 6 shows a performance comparison for the massive MIMO detection techniques with stair matrix initialization at -5 dB, 0 dB, 5 dB, 10 dB, and 20 dB SNR values. The massive detection techniques are simulated using Quadriga channels with 8×64 system size. The DL algorithm initialized achieves better performance than the V-BLAST algorithms at the cost of higher computational complexity. For example, the MMNet algorithm initialized with approximate matrix inversion methods attains an SER performance of 10^{-4} at

Massive MIMO detector	SER (-5dB)	SER (0dB)	SER (5dB)	SER (10dB)	SER (20dB)
MMNet-SOR	1.6×10^{-1}	6.1×10^{-2}	8.8×10^{-3}	2×10^{-3}	$7.7 imes 10^{-4}$
MMNet-SOR stair matrix	1×10^{-1}	4×10^{-2}	$5.4 imes 10^{-3}$	1×10^{-3}	$3.5 imes 10^{-4}$
MMNet-GS	1.2×10^{-1}	6×10^{-2}	$4.6 imes 10^{-3}$	9×10^{-4}	$2.3 imes 10^{-4}$
MMNet-GS stair matrix	1.1×10^{-1}	3.9×10^{-2}	4.3×10^{-3}	8×10^{-4}	$1.6 imes 10^{-4}$
ZF V-BLAST NS	$5.2 imes 10^{-2}$	4.3×10^{-2}	$4.1 imes 10^{-2}$	$3.3 imes 10^{-2}$	$2.6 imes 10^{-2}$
ZF V-BLAST NS stair matrix	$9.3 imes 10^{-3}$	$5.5 imes 10^{-3}$	$4.5 imes 10^{-3}$	4×10^{-3}	$3.7 imes 10^{-3}$
ZF V-BLAST NI	$4.3 imes 10^{-2}$	$3.5 imes 10^{-2}$	$3.3 imes 10^{-2}$	$2.7 imes 10^{-2}$	1.4×10^{-2}
ZF V-BLAST NI stair matrix	$9.3 imes 10^{-3}$	4.9×10^{-3}	4.3×10^{-3}	3.4×10^{-3}	$1.8 imes 10^{-3}$

TABLE 6. SER performance comparison for the proposed detection techniques with stair matrix initialization at -5 dB, 0 dB, 5 dB, 10dB, and 20dB SNR values using Quadriga channel models in an 8 × 64 massive MIMO system.

(a) SER vs. SNR of different DL detection methods with stair matrix initialization on Quadriga channels using offline training.

(b) Performance comparison for the proposed MMNet-GS detector with stair matrix initialization using Gaussian and Quadriga channels.

FIGURE 6. Performance of different massive MIMO DL detection techniques.

20 dB. In comparison, the ZF V-BLAST algorithms initialized with approximate matrix inversion methods achieve an SER performance of 10^{-2} and 10^{-3} .

The massive MIMO detectors initialized with a stair matrix have significantly better performance than detectors initialized with approximate matrix inversion methods that use a diagonal matrix. For instance, at 10 dB ZF V-BLAST NI with stair matrix has an SER of 3.4×10^{-3} while ZF V-BLAST NI with a diagonal matrix has a 2.7×10^{-2} . Moreover, the use of the stair matrix provides an SER performance improvement with negligible added computational complexity. It should be noted that ZF V-BLAST NS and ZF V-BLAST NI algorithms have reliable performance with significantly lower complexity than the DL algorithms. Moreover, DL detectors with stair matrix initialization, such as MMNet-GS and MMNet-SOR, performed better than the same detector initialized with the diagonal matrix. Additionally, a low number of linear iterations is utilized to achieve satisfactory performance with stair matrix initialization. Hence, the computational complexity of the MMNet-GS and MMNet-SOR detection techniques is reduced.

B. THE PROPOSED MASSIVE MIMO DETECTION TECHNIQUES WITH IMPERFECT CSI

In this section, instead of assuming the H to be perfectly known on the receiver side, we evaluate the proposed V-BLAST algorithms with imperfect channel estimation. The proposed V-BLAST algorithms use the estimated CSI for the Gramian matrix of the ZF linear detector such that B_{ZF} = $\hat{H}^{T}\hat{H}$. In Fig. 5, the proposed detectors which had the best performance in the scenario, i.e., ZF V-BLAST NS and ZF V-BLAST NI with stair matrix initialization, are tested with imperfect CSI using $r_0 = 0.99$ at the receiver. As shown in Fig. 7, the ZF V-BLAST NI was slightly affected by the imperfect channel estimation in the massive MIMO systems. Moreover, the technique attained satisfactory performance even with imperfect CSI, and channel correlation. It is worth noting that the ZF V-BLAST NS was more affected by imperfect channel information and the ZF V-BLAST NI algorithm outperformed ZF V-BLAST NS with both true channel and imperfect channel estimation scenarios.

In Fig. 8, we compare the performance of the ZF V-BLAST NS and ZF V-BLAST NI with stair matrix initialization using Gaussian channels and imperfect estimation of CSI as well as realistic channels in an 8×64 massive MIMO system. Similar results were obtained from the simulation in the

FIGURE 7. Performance comparison for different ZF V-BLAST algorithms with stair matrix initialization using 8×32 MIMO system under imperfect CSI.

FIGURE 8. Performance comparison for different ZF V-BLAST algorithms with stair matrix initialization using 8 \times 64 MIMO system under imperfect CSI and Quadriga channels.

 4×32 massive MIMO system. The proposed algorithms' performance slightly degrades in case of imperfect CSI compared to Gaussian channel models. Moreover, it was shown that the most challenging scenario was using the realistic channels simulated using the Quadriga model. The proposed ZF V-BLAST NI with stair matrix initialization algorithm achieved good performance in realistic channels, followed by ZF V-BLAST NS with stair matrix initialization which achieved satisfactory results in the same scenario.

VII. CONCLUSION

In this paper, several low complexity massive MIMO detectors have been proposed based on approximate matrix inversion methods, stair matrix, DL approach, and V-BLAST. Proposed detectors have shown a significant performance gain and a remarkable complexity reduction. It is worthnoting that the NS and NI employment with ZF V-BLAST and stair matrix lead to a notable performance improvement while maintaining low computational complexity. In order to avoid any misleading conclusion, realistic channel scenario is considered and proposed detectors also achieved a satisfactory balance between performance gain and complexity reduction.

VIII. FUTURE RESEARCH DIRECTIONS

For future studies, different approximate matrix inverse techniques can be utilized in massive MIMO detection algorithms where exact matrix inversion is a computationally expensive task. Stair matrix initialization can be exploited as well to enhance approximate matrix inversion methods that are employed in various fields, such as machine learning and communication systems. For example, the approximate methods with the stair matrix can be used in various optimization problems in massive MIMO detection, beamforming, precoding, and channel estimation. In massive MIMO detection, the stair matrix can improve the performance and the convergence rate of linear iterative detectors while requiring a lower number of iterations. Accordingly, the stair matrix implementation can minimize the complexity of approximate matrix inversion methods without sacrificing performance for the same massive MIMO system.

Several parameters can be investigated to optimize the performance and reduce the computational cost in the neural network architecture of the DL-based massive MIMO detection techniques. Furthermore, the development of new machine learning and artificial intelligence algorithms specifically tailored for the unique characteristics of massive MIMO systems can be explored. The stair matrix can be used instead of the diagonal matrix with approximate matrix inversion methods to enhance efficiency in DL-based detectors. In addition, the effectiveness of the detection techniques using linear iterative approximate matrix inversion methods and DL can be studied for cell-free massive MIMO systems. The linear iterative and DL-based detectors have the potential to be adapted to the frequency selective channel, thereby opening up the possibility of their implementation in millimeter-wave (mmWave) massive MIMO systems. Moreover, approximate matrix inversion methods and DL can be employed in various fields to enhance performance and reduce complexity for real-time applications. Based on the channel hardening phenomenon, we believe that this work can be extended using different matrix structures such as the banded matrix, the Hankel matrix, and even the scaled identity matrix. Other initialization schemes can be exploited as well. For instance, the first iteration of NI can be exploited to initially approximate the matrix inversion.

REFERENCES

- M. A. Albreem, M. Juntti, and S. Shahabuddin, "Massive MIMO detection techniques: A survey," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 4, pp. 3109–3132, 4th Quart., 2019.
- [2] M. A. Albreem, A. H. Alhabbash, S. Shahabuddin, and M. Juntti, "Deep learning for massive MIMO uplink detectors," *IEEE Commun. Surveys Tuts.*, vol. 24, no. 1, pp. 741–766, 1st Quart., 2022.
- [3] J.-C. Chen, "A low complexity data detection algorithm for uplink multiuser massive MIMO systems," *IEEE J. Sel. Areas Commun.*, vol. 35, no. 8, pp. 1701–1714, Aug. 2017.
- [4] Y.-G. Lim, C.-B. Chae, and G. Caire, "Performance analysis of massive MIMO for cell-boundary users," *IEEE Trans. Wireless Commun.*, vol. 14, no. 12, pp. 6827–6842, Dec. 2015.
- [5] D. J. Costello, "Fundamentals of wireless communication (TSE, D. and Viswanath, P.)[book review]," *IEEE Trans. Inf. Theory*, vol. 55, no. 2, pp. 919–920, Feb. 2009.
- [6] J. Sun, Y. Zhang, J. Xue, and Z. Xu, "Learning to search for MIMO detection," *IEEE Trans. Wireless Commun.*, vol. 19, no. 11, pp. 7571–7584, Nov. 2020.

- [7] M. A. Albreem, M. H. Alsharif, and S. Kim, "A low complexity nearoptimal iterative linear detector for massive MIMO in realistic radio channels of 5G communication systems," *Entropy*, vol. 22, no. 4, p. 388, Mar. 2020.
- [8] H. Prabhu, J. Rodrigues, O. Edfors, and F. Rusek, "Approximative matrix inverse computations for very-large MIMO and applications to linear precoding systems," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, Apr. 2013, pp. 2710–2715.
- [9] L. Fang, L. Xu, and D. D. Huang, "Low complexity iterative MMSE-PIC detection for medium-size massive MIMO," *IEEE Wireless Commun. Lett.*, vol. 5, no. 1, pp. 108–111, Feb. 2016.
- [10] C. Tang, C. Liu, L. Yuan, and Z. Xing, "High precision low complexity matrix inversion based on Newton iteration for data detection in the massive MIMO," *IEEE Commun. Lett.*, vol. 20, no. 3, pp. 490–493, Mar. 2016.
- [11] H. Lu, "Stair matrices and their generalizations with applications to iterative methods I: A generalization of the successive overrelaxation method," *SIAM J. Numer. Anal.*, vol. 37, no. 1, pp. 1–17, Jan. 1999.
- [12] M. Albreem, M. Juntti, and S. Shahabuddin, "Efficient initialisation of iterative linear massive MIMO detectors using a stair matrix," *Electron. Lett.*, vol. 56, no. 1, pp. 50–52, Jan. 2020.
- [13] D. Tse and P. Viswanath, Fundamentals of wireless communication. Cambridge, U.K.: Cambridge Univ. Press, 2005.
- [14] D. Jakhar and I. Kaur, "Artificial intelligence, machine learning and deep learning: Definitions and differences," *Clin. Experim. Dermatol.*, vol. 45, no. 1, pp. 131–132, Jan. 2020.
- [15] L. V. Nguyen, N. T. Nguyen, N. H. Tran, M. Juntti, A. L. Swindlehurst, and D. H. N. Nguyen, "Leveraging deep neural networks for massive MIMO data detection," *IEEE Wireless Commun.*, vol. 30, no. 1, pp. 174–180, Feb. 2023.
- [16] S. Kumar, R. Mahapatra, P. Kumar, and A. Singh, "Decoder design for massive-MIMO systems using deep learning," *IEEE Syst. J.*, vol. 16, no. 4, pp. 6614–6623, Dec. 2022.
- [17] S. Kumar, A. Singh, and R. Mahapatra, "Deep learning based massive-MIMO decoder," in *Proc. IEEE Int. Conf. Adv. Netw. Telecommun. Syst.* (ANTS), Dec. 2019, pp. 1–6.
- [18] J. Li, Q. Zhang, X. Xin, Y. Tao, Q. Tian, F. Tian, D. Chen, Y. Shen, G. Cao, Z. Gao, and J. Qian, "Deep learning-based massive MIMO CSI feedback," in *Proc. 18th Int. Conf. Opt. Commun. Netw. (ICOCN)*, Aug. 2019, pp. 1–3.
- [19] M. Chen, J. Guo, C.-K. Wen, S. Jin, G. Y. Li, and A. Yang, "Deep learningbased implicit CSI feedback in massive MIMO," *IEEE Trans. Commun.*, vol. 70, no. 2, pp. 935–950, Feb. 2022.
- [20] N. T. Nguyen, K. Lee, and H. DaiIEEE, "Application of deep learning to sphere decoding for large MIMO systems," *IEEE Trans. Wireless Commun.*, vol. 20, no. 10, pp. 6787–6803, Oct. 2021.
- [21] S. Kumar, A. Singh, and R. Mahapatra, "DLNet: Deep learning-aided massive MIMO decoder," AEU Int. J. Electron. Commun., vol. 155, Oct. 2022, Art. no. 154350.
- [22] N. Samuel, T. Diskin, and A. Wiesel, "Learning to detect," *IEEE Trans. Signal Process.*, vol. 67, no. 10, pp. 2554–2564, May 2019.
- [23] H. He, C.-K. Wen, S. Jin, and G. Y. Li, "A model-driven deep learning network for MIMO detection," in *Proc. IEEE Global Conf. Signal Inf. Process. (GlobalSIP)*, Nov. 2018, pp. 584–588.
- [24] H. He, C.-K. Wen, S. Jin, and G. Y. Li, "Model-driven deep learning for MIMO detection," *IEEE Trans. Signal Process.*, vol. 68, pp. 1702–1715, 2020.
- [25] M. Khani, M. Alizadeh, J. Hoydis, and P. Fleming, "Adaptive neural signal detection for massive MIMO," *IEEE Trans. Wireless Commun.*, vol. 19, no. 8, pp. 5635–5648, Aug. 2020.
- [26] M. Goutay, F. A. Aoudia, and J. Hoydis, "Deep HyperNetwork-based MIMO detection," in *Proc. IEEE 21st Int. Workshop Signal Process. Adv. Wireless Commun. (SPAWC)*, May 2020, pp. 1–5.
- [27] Y. Yu, J. Ying, P. Wang, and L. Guo, "A data-driven deep learning network for massive MIMO detection with high-order QAM," *J. Commun. Netw.*, vol. 25, no. 1, pp. 50–60, Feb. 2023.
- [28] L. V. Nguyen, D. H. N. Nguyen, and A. L. Swindlehurst, "Deep learning for estimation and pilot signal design in few-bit massive MIMO systems," *IEEE Trans. Wireless Commun.*, vol. 22, no. 1, pp. 379–392, Jan. 2023.
- [29] K. He, Z. Wang, D. Li, F. Zhu, and L. Fan, "Ultra-reliable MU-MIMO detector based on deep learning for 5G/B5G-enabled IoT," *Phys. Commun.*, vol. 43, Dec. 2020, Art. no. 101181.
- [30] S. N. Karahan and A. Kalaycioğlu, "Utilization of deep learning architectures for MIMO detection," *Commun. Fac. Sci. Univ. Ankara Ser. A2–A3 Phys. Sci. Eng.*, vol. 64, no. 2, pp. 81–94, Dec. 2022.

- [31] A. J. Almasadeh, K. A. Alnajjar, and M. A. Albreem, "Enhanced deep learning for massive MIMO detection using approximate matrix inversion," in *Proc. 5th Int. Conf. Commun., Signal Process., Their Appl.* (ICCSPA), Dec. 2022, pp. 1–6.
- [32] K. A. Alnajjar, "Receiver design for massive MIMO," Tech. Rep., 2015.
- [33] K. A. Alnajjar and S. Abdallah, "Performance of low complexity receivers for massive MIMO with channel estimation and correlation," in *Proc. IEEE 3rd Int. Symp. Telecommun. Technol. (ISTT)*, Nov. 2016, pp. 1–5.
- [34] S. Jaeckel, L. Raschkowski, K. Börner, and L. Thiele, "QuaDRiGa: A 3-D multi-cell channel model with time evolution for enabling virtual field trials," *IEEE Trans. Antennas Propag.*, vol. 62, no. 6, pp. 3242–3256, Jun. 2014.
- [35] C. Wang, E. K. S. Au, R. D. Murch, and V. K. N. Lau, "Closed-form outage probability and BER of MIMO zero-forcing receiver in the presence of imperfect CSI," in *Proc. IEEE 7th Workshop Signal Process. Adv. Wireless Commun.*, Jul. 2006, pp. 1–5.
- [36] B. Trotobas, A. Nafkha, and Y. Louët, "A review to massive MIMO detection algorithms: Theory and implementation," in Advanced Radio Frequency Antennas for Modern Communication and Medical Systems. London, U.K.: IntechOpen, 2020.
- [37] S. Yang and L. Hanzo, "Fifty years of MIMO detection: The road to large-scale MIMOs," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 1941–1988, 4th Quart., 2015.
- [38] M. Ylinen, A. Burian, and J. Takala, "Direct versus iterative methods for fixed-point implementation of matrix inversion," in *Proc. IEEE Int. Symp. Circuits Syst.*, May 2004, p. 225.
- [39] Y. Lee and S.-I. Sou, "On improving Gauss–Seidel iteration for signal detection in uplink multiuser massive MIMO systems," in *Proc. 3rd Int. Conf. Comput. Commun. Syst. (ICCCS)*, Apr. 2018, pp. 268–272.
- [40] Z. Wu, C. Zhang, Y. Xue, S. Xu, and X. You, "Efficient architecture for soft-output massive MIMO detection with Gauss–Seidel method," in *Proc. IEEE Int. Symp. Circuits Syst. (ISCAS)*, May 2016, pp. 1886–1889.
- [41] Z. Wu, Y. Xue, X. You, and C. Zhang, "Hardware efficient detection for massive MIMO uplink with parallel Gauss–Seidel method," in *Proc. 22nd Int. Conf. Digit. Signal Process. (DSP)*, Aug. 2017, pp. 1–5.
- [42] X. Gao, L. Dai, Y. Hu, Z. Wang, and Z. Wang, "Matrix inversion-less signal detection using SOR method for uplink large-scale MIMO systems," in *Proc. IEEE Global Commun. Conf.*, Dec. 2014, pp. 3291–3295.
- [43] P. Zhang, L. Liu, G. Peng, and S. Wei, "Large-scale MIMO detection design and FPGA implementations using SOR method," in *Proc. 8th IEEE Int. Conf. Commun. Softw. Netw. (ICCSN)*, Jun. 2016, pp. 206–210.
- [44] F. Jiang, C. Li, Z. Gong, and R. Su, "Stair matrix and its applications to massive MIMO uplink data detection," *IEEE Trans. Commun.*, vol. 66, no. 6, pp. 2437–2455, Jun. 2018.
- [45] L. Dai, X. Gao, X. Su, S. Han, C.-L. I, and Z. Wang, "Low-complexity softoutput signal detection based on Gauss–Seidel method for uplink multiuser large-scale MIMO systems," *IEEE Trans. Veh. Technol.*, vol. 64, no. 10, pp. 4839–4845, Oct. 2015.
- [46] B. Yin, M. Wu, J. R. Cavallaro, and C. Studer, "Conjugate gradient-based soft-output detection and precoding in massive MIMO systems," in *Proc. IEEE Global Commun. Conf.*, Dec. 2014, pp. 3696–3701.
- [47] Z. Albataineh, "Iterative signal detection based on MSD-CG method for uplink massive MIMO systems," in *Proc. 16th Int. Multi-Conf. Syst.*, *Signals Devices (SSD)*, Mar. 2019, pp. 539–544.
- [48] A. Chockalingam, "Low-complexity algorithms for large-MIMO detection," in *Proc. 4th Int. Symp. Commun., Control Signal Process. (ISCCSP)*, Mar. 2010, pp. 1–6.
- [49] W. Fukuda, T. Abiko, T. Nishimura, T. Ohgane, Y. Ogawa, Y. Ohwatari, and Y. Kishiyama, "Low-complexity detection based on belief propagation in a massive MIMO system," in *Proc. IEEE 77th Veh. Technol. Conf. (VTC Spring)*, Jun. 2013, pp. 1–5.
- [50] M. Wu, C. Dick, J. R. Cavallaro, and C. Studer, "High-throughput data detection for massive MU-MIMO-OFDM using coordinate descent," *IEEE Trans. Circuits Syst. I, Reg. Papers*, vol. 63, no. 12, pp. 2357–2367, Dec. 2016.
- [51] S. Shahabuddin, M. Juntti, and C. Studer, "ADMM-based infinity norm detection for large MU-MIMO: Algorithm and VLSI architecture," in *Proc. IEEE Int. Symp. Circuits Syst. (ISCAS)*, May 2017, pp. 1–4.
- [52] K. A. Alnajjar, P. J. Smith, and G. K. Woodward, "Performance of massive MIMO V-BLAST with channel correlation and imperfect CSI," in *Proc. Australas. Telecommun. Netw. Appl. Conf. (ATNAC)*, Nov. 2014, pp. 170–175.

- [53] K. A. Alnajjar, P. J. Smith, and G. K. Woodward, "Low complexity V-BLAST for massive MIMO," in *Proc. Austral. Commun. Theory Workshop* (AusCTW), Feb. 2014, pp. 22–26.
- [54] H. Schwerdtfeger, Introduction to Linear Algebra and the Theory of Matrices. Winschoterdiep, The Netherlands: P. Noordhoff, 1961.
- [55] J. H. Chong, C. K. Ng, N. K. Noordin, and B. M. Ali, "A low computational complexity V-BLAST/STBC detection mechanism in MIMO system," *Hum.-centric Comput. Inf. Sci.*, vol. 4, no. 1, pp. 1–28, Dec. 2014.
- [56] R. Hunger, "Floating point operations matrix-vector calculus, volume 2019," Munich Univ. Technol., Inst. Circuit Theory Signal, Munich, Germany, Tech. Rep., 2005.
- [57] K. A. Alnajjar, P. J. Smith, G. K. Woodward, and D. A. Basnayaka, "Design and analysis of a reduced complexity MRC V-BLAST receiver for massive MIMO," in *Proc. IEEE 17th Int. Workshop Signal Process. Adv. Wireless Commun. (SPAWC)*, Jul. 2016, pp. 1–5.

KHAWLA A. ALNAJJAR (Member, IEEE) received the B.S. degree in electrical engineering and communication track from United Arab Emirates University (UAEU), Al Ain, in 2008, the M.S. and P.E.E. degrees in electrical engineering from Columbia University, New York, in 2010 and 2012, respectively, and the Ph.D. degree in electrical and electronics engineering from the University of Canterbury, Christchurch, New Zealand, in 2015. She is currently an Assistant

Professor with the Department of Electrical and Computer Engineering, University of Sharjah, United Arab Emirates. Her research interests include wireless communication systems, mathematical statistics, network information theory, and power grids. She has received more than 30 competitive awards for her successful studies and research during these ten years.

MAHMOUD A. ALBREEM (Senior Member, IEEE) received the B.Eng. degree in electrical engineering (EE) from the Islamic University of Gaza, Palestine, in 2008, and the M.Sc. and Ph.D. degrees in EE from Universiti Sains Malaysia (USM), Malaysia, in 2010 and 2013, respectively. From 2014 to 2016, he was a Senior Lecturer with Universiti Malaysia Perlis. In 2016 to 2021, he chaired with the Department of Electronics and Communications Engineering, A'Sharqiyah Uni-

versity, Oman. Currently, he is an Assistant Professor with the Department of Electrical Engineering, University of Sharjah, UAE. He is also a Visiting Assistant Professor with Centre for Wireless Communications (CWC), University of Oulu, Finland. He is the author of more than 80 journal and conference papers. His research interests include multiple-input multiple-output (MIMO) detection and precoding techniques, machine learning applications for wireless communication systems, and green communications. He received several scholarships and grants, such as Nokia Foundation Centennial Grant (2018), USM Fellowship (2011–2013), and Best Master's Thesis Award of the School of Electrical and Electronics Engineering, USM (2010). He served on the Editorial Board for Journal of Wireless Communications and Mobile Computing.

ALI J. ALMASADEH received the bachelor's degree in electrical engineering from the University of Jordan, Amman, Jordan, in 2015. He is currently pursuing the M.Sc. degree in electrical and electronics engineering with the University of Sharjah. He was with telecommunication industry in both mobile access networks and transport networks development with Etisalat, for eight years. His research interests include wireless communications, artificial intelligence, machine learning, and signal detection.

...